Title: Effect of 50% ethanolic extract of Syzygium aromaticum (L.) Merr. & Perry. (clove) on Sexual behaviour of normal male rats.

Authors:

Tajuddin D Tajuddin (tajuddin_amu@rediffmail.com)
Shamshad D Ahmad (shamshad_2003@rediffmail.com)
Abdul D Latif (latifabdul@rediffmail.com)
Iqbal D Ahmad (qasimiqbal@rediffmail.com)

Version: 5 Date: 29 September 2004

Author's response to reviews: see over
Title of manuscript: Effect of 50% ethanolic extract of *Syzygium aromaticum* (L.) Merr. & Perry. (clove) on sexual behaviour of normal male rats

MS ID: 3430433273709239

Journal: BMC Complementary and Alternative Medicine

The changes have been made in the manuscript in accordance with the suggestions made by Mr. Srinivas NAMMI in his 2nd review report. A point-by-point response to the comments is provided herewith.

**Minor Essential Revisions**

1. The reference for the effect of spices on gastric mucosa has now been mentioned in the discussion section of the manuscript.

2. In Table 1 the symbol @ used for indicating the time period has now been deleted. Instead of this time period written in the first column of the corresponding parameters as suggested. The symbols # and ## for results with no significance have also been deleted and instead of these symbols only NS is used.

3. All the titles of the references bolded. And all journal titles italicized as per the format of the journal.

4. Italicization of the first level heading: Materials and methods has been removed.

5. We have thoroughly checked the manuscript again for typographical and grammatical errors and corrections are made wherever required.